Response to Reviewer 1 Comments

We thank the reviewer for the time and effort invested into reviewing our manuscript. Here are our response to your comments and feedback:

Point 1: Mechanics of the writing The writing mechanic is full of grammatical and typographical errors. Below, please find some examples from the manuscript
Response 1: We have revised theenglish and grammar.
added the results and p-values to the comparison variables. Thank you for this suggestion.

Point 2: Why was transverse colostomy was performed?
Response 2: Due to ischemic change of sigmoid colon, we performed transverse colostomy.

Point 3: Wide type? Did you mean KRAS Wild Type.
Response 3: We mean KRAS Wild Type.